Author's response to reviews

Title: Bladder augmentation with continent stoma: a surgical option by association between Mitrofanoff's principle and Monti technique.

Authors:

Marcelo F Cassini (marcelo.cassini@globo.com)
Antonio A Rodrigues Jr (aantunesrj@yahoo.com.br)
Silvio Tucci Jr (stuccijr@yahoo.com.br)
Adauto J Cologna (ajcologn@fmrp.usp.br)
Rodolfo B Reis (rbreis@keynet.com.br)
Antonio CP Martins (antonio@capovox.com.br)
Haylton J Suaid (hjsuaid@fmrp.usp.br)

Version: 4 Date: 19 June 2010

Author's response to reviews:

Dear responsible,

I made the suggested changes, restructuring the abstract into one column,...

Thank you very much for attention,

Marcelo F Cassini